Author's response to reviews

Title: Treatment for femoral head migration of a hemiarthroplasty that was performed 22 years previously in a patient at the age of 20 years: a case report

Authors:

Akio Kanda (tetsutakan@aol.com)
Kazuo Kaneko (hobata@juntendo.ac.jp)
Osamu Obayashi (obys@mx7.ttcn.ne.jp)
Atsuhiyo Mogami (moga13@mx8.ttcn.ne.jp)

Version: 3
Date: 13 September 2014

Author's response to reviews: see over
Dear Editor

Thank you for your letter of September 8, 2014. We have carefully examined the reviewers' comments and are submitting a revised manuscript. I would like to submit the enclosed paper entitled “Treatment for femoral head migration of a hemiarthroplasty that was performed 22 years previously in a patient at the age of 20 years: a case report (Journal of Medical Case Reports MS: 1955672938139736) “ to your journal for possible publication. We believe it is the first report of this kind in the literature because of an unexpected event in the course of treating a patient. And it will significantly advance our understanding of a particular disease etiology or mechanism.

And it is an original case report of interest to a particular clinical speciality of medicine. The paper is submitted to be considered for publication as original articles in your journal. This manuscript has not been submitted simultaneously elsewhere. The responsible author (Akio Kanda) certifies that all the co-authors (K.keneko, O obayashi, A. Mogami) have seen and agreed to the contents of the manuscript.

I hope you will find this manuscript acceptable for publication in your journal.